

Welch, Vicki and Fowler, Nadine and Ross, Ewan and Withington, Richard and McGhee, Kenny (2018) In and Beyond the Care Setting : Relationships Between Young People and Care Workers: A Literature Review. [Report],

This version is available at https://strathprints.strath.ac.uk/65636/

Strathprints is designed to allow users to access the research output of the University of Strathclyde. Unless otherwise explicitly stated on the manuscript, Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Please check the manuscript for details of any other licences that may have been applied. You may not engage in further distribution of the material for any profitmaking activities or any commercial gain. You may freely distribute both the url (<u>https://strathprints.strath.ac.uk/</u>) and the content of this paper for research or private study, educational, or not-for-profit purposes without prior permission or charge.

Any correspondence concerning this service should be sent to the Strathprints administrator: strathprints@strath.ac.uk

The Strathprints institutional repository (https://strathprints.strath.ac.uk) is a digital archive of University of Strathclyde research outputs. It has been developed to disseminate open access research outputs, expose data about those outputs, and enable the management and persistent access to Strathclyde's intellectual output.



In and beyond the care setting: relationships between young people and care workers

A literature review

CELCIS: Vicki Welch, Nadine Fowler, Ewan Ross, Richard Withington, Kenny McGhee.



Acknowledgements

This literature review was possible because of funding from Care Visions, linked to the evaluation of their 'Why Not' scheme. We are very grateful for this support, enabling us to make learning available to others working with care experienced young people. We would like to thank all those who helped to identify or access the sources included in this review.

Table of Contents

Acknowledgements	i
Introduction and overview	3
Review methodology	3
Transition from care: background and policy context	4
Findings: Part 1. Leaving care Part 2. The importance of relationships Part 3. Themes around relationships Part 4. Relationships in and beyond the care setting	7 9
Summary and conclusion	21
References	25

Introduction and overview

This review seeks to identify and summarise findings from literature about the nature of relationships that develop between older children and young people, and those caring for them within and beyond residential and fostering settings. We make particular efforts to include studies that gather the views of young people themselves. We consider the issues and challenges that young people face in moving on from care, the type of support they receive during this process, and focus on the relational elements of this support. The study as a whole focuses on young people in adolescence as they approach the point where they will leave care and undertake the transition towards more independent living.

The review will briefly outline the background and policy context, before discussing some of the key themes found in the literature.

- Part 1 explores the issues facing young people leaving care and the need for more targeted, specialist support.
- Part 2 discusses the importance of relationships for young people leaving care and the growing interest in relationship-based practice.
- Part 3 explores various aspects of relationships valued by care leavers.
- Part 4 looks at relationships in different types of care settings.

Review methodology

Given the breadth of the topic, the review needed to follow a broadly inclusive approach. We developed a systematic search strategy for two electronic databases: *Scopus* and *Social Care Online*. These databases were chosen to give a breadth of coverage of peer reviewed material and other sources. In addition to these searches, we identified key grey literature and approached various colleagues asking them to recommend relevant sources. Finally, we hand-searched reference lists to identify additional material not found through other methods.

Although we conducted searches without restricting date periods, we have focussed our efforts mainly on material published since 2000 in order to ensure the review's contemporary relevance. However, where we consider older works to have been highly influential and relevant to the field, we include them.

Example of search term used in Scopus

TITLE(relation* or befriend* or mentor* or buddy* or interperson* or friendship*) AND TITLE-ABS-KEY(client* or user* or "young pe*" or youth* or adolescent* or "care leaver*" or "care experience*") AND TITLE-ABS-KEY(worker* or practitioner* or carer* or professional* or teacher* or tutor* or "pedagog*") AND TITLE ("after care*" or aftercare* or "through care*" or throughcare* or "looked after*" or "looked-after" or "in care*" or "state care*" or "foster care*" or "residential care*" or transition* or "moral adoption" or befriend* or "life coach*") AND SUBJAREA (mult OR medi OR nurs OR vete OR dent OR heal OR mult OR arts OR busi OR deci OR econ OR psyc OR soci) A total of 145 returns were obtained from the structured searches, including some duplicated sources. We then developed further inclusion and exclusion criteria to identify those works that were most directly relevant to the topic. The following criteria were used:

- Include published and 'grey literature' from peer reviewed and other sources
- Include material related to supportive relationships with adolescent youth in foster or residential care
- Include findings from good quality literature reviews where available
- Include material focused on Scotland, UK and, where relevant, other geographies
- Include material that directly sought care leavers' experiences of the process of leaving care and relational elements of this
- Exclude material not available in English
- Exclude material where it was not possible to distinguish views and issues for care leavers or those preparing to leave care from the views of other participants.

After removing duplicates, we scrutinised the abstracts and/or introductory information of each study to ensure fit with the inclusion and exclusion criteria. Full text documents were obtained for all relevant works. A total of 104 reviewed works were included in the next stage, during which we systematically extracted and recorded relevant information from each text; most of these sources included something of value, and were included in the report in some way. The extracted information was then organised and reported thematically in our Findings section.

For context, we also consulted some additional sources, including relevant statistics and government guidance. In total, we reference 119 documents in this literature review. Before discussing the reviewed materials, we provide contextual information about leaving care in Scotland.

Transition from care: background and policy context

Children and young people become looked after for a variety of reasons, including abuse, neglect, and involvement in offending behaviour. They can be placed away from home as a condition of a compulsory supervision order made through the Children's Hearing System, or less often a court. Alternatively, children and young people may be looked after on a voluntary basis as laid out in S25 of the Children (Scotland) Act 1995.

In Scotland, children and young people who are cared from away from home will reside in one of several placement types: foster care; kinship care; residential home; residential school or secure care. On 31 July 2016, 15,317 children were looked after in Scotland (Scottish Government, 2017). Of that total, 35% were in foster care, 10% in residential care and 26% in kinship care, and a quarter were being looked after at home under supervision. Of the 1,477 in residential care, 39% were in local authority children's units, 26% were in residential schools, and 4% were in secure care, with the remaining 31% in other types of residential care such as a non-local authority home or specialist provision for children with complex needs (Scottish Government, 2017).

Young people looked after on or after their 16th birthday are eligible to receive aftercare support when moving on from care. As long as they continue to have eligible needs, young people moving on from care can now receive support up to their 26th birthday, as set out in the Children and Young People (Scotland) Act 2014. The provision of Throughcare and Aftercare support is set out in the Children (Scotland) Act 1995 and is further defined by the Leaving Care Regulations 2004 (Scottish Executive, 2004). This outlines the role of social work services in making support available to young people moving on from care is defined.

As well as legislation, a succession of reports since the 1990s have emphasised the need for the provision of more coherent and robust support for young people leaving the care system. Calls for these young people to have continuing supportive relationships led to developments such as the *Staying Put Scotland* guidance (Scottish Government, 2013), and the aforementioned extension of aftercare support to the age of 26 in the Children and Young People (Scotland) Act 2014. Research suggests that the single most influential factor in improving outcomes for looked after young people and care leavers is the age at which they transition from care to independence (Broad, 2007; Stein, 2012; Wade, 1997). It is therefore significant that the 2014 Act also sets out new conditions, allowing young people to remain in stable care placements up to the age of 21.

It is also important to understand the concept of 'corporate parenting' in relation to moving on from care. This term describes collective action to improve the life chances of looked after children and care leavers, whereby the state (as a whole and in its constituent parts) seeks to mitigate future disadvantage by replicating the kind of support that non-looked after young people may expect from a parent or parents (Scottish Government, 2008). The Government has recently consolidated this concept, and it is more clearly defined in guidance and statute related to the 2014 Act.

Over time, developments such as those described above have been driving a renewed interest in relationship-based practice. This is exemplified by the prominence of guidance advocating the continuity of supportive relationships, such as *Staying Put Scotland* (Scottish Government, 2013), and the new *Keep on Caring* cross-government strategy in England (Department for Education et al., 2016).

Findings:

Part 1. Leaving care

There have been a number of UK studies over the past 30 years that have focused on young people leaving the care system (Stein, 2004). This research highlights that care leavers are a very diverse group, with a wide range of reasons for entering care and diverse experiences within the care system. Dixon & Stein point out that:

Some young people will have positive and valued experiences of being looked after, while for others, it may generate further problems [...] they may experience further disruption through placement movement, the erosion of family and community links, and the failure to have their basic developmental, educational and health needs met (2005, p. 10).

The majority of young people have historically left care at 16 or 17 years old. At this point, the state was widely considered no longer responsible, legally or morally, for providing care. Several studies highlight that this contrasts a growing tendency for most young people in the wider population to remain at home well into their twenties, and for the transition to independent living to be undertaken in stages (see for example, Bynner, Elias, McNight, Pan, & Pierre, 2002).

Characteristically, young care leavers will experience accelerated, compressed transitions (Biehal, Clayden, Stein, & Wade, 1995). The significant life changes that define the transition phase, such as leaving school, seeking employment, finding and adjusting to new accommodation, and becoming financially independent, tend to impact care leavers soon after moving on from their last care placement (Dixon & Stein, 2005). This means that care leavers generally assume a range of adult responsibilities at a much younger age than their peers (Biehal et al., 1995; Dixon & Stein, 2005; Pinkerton & McCrea, 1999).

Furthermore, in contrast to many young people on the journey towards independence, care leavers are often unable to rely on emotional, practical, or financial support from their families (Barnardo's, 2014; Biehal & Wade, 1996; Sinclair, Baker, Wilson, & Gibbs, 2005). Care leavers are also at a high risk of social exclusion, especially through homelessness, loneliness, isolation, unemployment, poverty, and mental ill health (Biehal et al., 1995; Broad, 1998; Stein & Carey, 1986). This evidence indicates that young people are often poorly equipped to cope with the significant challenges of life after the care setting.

As well as all of the practical difficulties that have to be overcome by young people in transition, studies report that insufficient attention is paid to the emotional aspects of moving towards greater independence, contributing to the poor outcomes that this group experience as a matter of course (Adley & Jupp Kina, 2014). After leaving care, many young people live alone. Often, this is not their choice, and many struggle with the social implications of doing so (Wade, 2008). Additionally, Marion, Paulsen and Goyette (2017, p. 578) argue that young people moving on from care had few relationships, resulting in a feeling of 'nobody to turn to'. Young care leavers are also more likely to enter into early parenthood or develop extended caretaking responsibilities as a result of ambiguous loss of family (Lee, Cole, & Munson, 2016).

Typically, care leavers face a wide variety of challenges that both pre-date their care experience, and result from it. They may continue to struggle with these issues throughout their lives (Duncalf, 2010, p. 42). The nature of these concerns and poor

outcomes for care leavers is widely documented (for example, Aldgate, 1994; Biehal et al., 1995; Stein & Carey, 1986). However, fewer sources describe how services should respond to these concerns; Wade and Munro point out that:

At present, we know much more about the problems and risks faced by young people leaving care (about what does not work well) than we do about the forms of support that may be effective in helping them to negotiate successful transitions into adulthood (2008, p. 219).

In Scotland, these concerns resulted in guidance requiring the formation and consolidation of Throughcare and Aftercare services and the introduction of Pathway Assessments and Plans as a means of personalising, and quality assuring, the support provided (Scottish Executive, 2004). This support included personal support, help with accommodation and housing, financial assistance, and help with careers and further education (Dixon & Stein, 2005). In an evaluation of four projects in England, Biehal and colleagues found evidence that support in these areas could contribute to better outcomes for care leavers, particularly given the very poor starting points that many experience (Biehal et al., 1995). Similarly, there is evidence from Scottish research that good preparation for leaving care can help young people to cope with the transition (Dixon & Stein, 2005). Ultimately, Coyle and Pinkerton observed that:

For too many years the phrase 'after care, an after-thought', summed up the lack of attention given to preparing young people in care for adult life and to the need for development of leaving-care and aftercare services. That stance has changed in the United Kingdom, however, and indeed has shifted positively and dramatically over the last decade (2012, p. 297).

Part 2. The importance of relationships

Where evidence does exist about how young people manage the transition from care, relationships feature strongly. This should not be a surprise:

For the majority of us, the quality of our relationships with other people remains the most important area of experience and the yardstick by which we measure happiness and contentment (Howe, 1995, p. 1).

Engaging with people through helping relationships has traditionally been a core function of social work and social care (Coady, 2014; Howe, 1995). Building and maintaining relationships with individuals (i.e. having the capacity to relate to others and their problems) is important in achieving successful outcomes. However, this skill is difficult to develop and exercise effectively, particularly when working with very challenging young people and their families (O'Leary, Tsui, & Ruch, 2013; Winter, 2015).

Nonetheless, access to supportive relationships is particularly important for care leavers, as Wade points out:

Leaving care is often a time of reappraisal. It is a time when relationships are re-evaluated to gauge the extent to which they can be relied upon in the future. At such a stage, it is important for young people to know, from the pool of people in their lives, who may be willing to provide practical and emotional support as they move forward into the adult world. It can be a time of considerable uncertainty (2008, p. 39).

The recent Care Inquiry, set up by a group of English charities, drew on evidence from participants with direct experience of care, or of working within the care system, to look at how best to provide stable, permanent homes for vulnerable children who were unable to live with their parents. In highlighting ways that the care system fails too many children, the Inquiry concluded that greater focus should be given to making and maintaining relationships:

Relationships are the golden thread in children's lives [...] the quality of a child's relationships is the lens through which we should view what we do and plan to do (Care Inquiry, 2013, p. 9).

The Barnardo's report, *Someone to Care*, presented findings from in-depth interviews with 62 young care leavers. This identified that young people felt they needed someone to care about them, someone to talk to, someone to be with, someone to set standards, and someone to show them the way (Barnardo's, 2014). These descriptions clearly indicate that those young people viewed close and trusted supportive relationships as a critical part of their care that was instrumental in helping them to make the transition.

In exploring the issues of relationships with mentors, young people have highlighted a number of important characteristics that they believe are important for developing these important, supportive relationships. Similar to the findings discussed by Barnardo's, these are described as availability, seeing the young person as positive and trustworthy, offering 'parental gestures', continued transitional support, support above and beyond paid duties, feeling valued, and being there for the long-term (Newton, Harris, Hubbard, & Craig, 2017; Sulimani-Aidan, 2017).

The ascendance of managerialism and the extension of regulatory approaches to services in the 1980's and 90's, led to an increased focus on outcomes and targets in relation to young people in residential care (Coady, 2014; Steckley & Smith, 2011). It is argued by some commentators that in seeking to define, control, and police boundaries between professionals and their clients, managerial approaches have often down-played the importance of nurturing, and developing the close and trusting relationships that young people clearly require (Smith, 2009). For instance, Moore, McArthur, Death, Tilbury, and Roche (2018) suggest that managerialism, in the form of wider organisational and structural factors, impedes on the ability of children and young people to form relationships with staff members that are safe, trustworthy and reliable.

There has been a growing interest in relationship-based practice throughout various strands of social work in recent years. This interest recognises that case management

approaches which focus only on standards, outcomes, and targets, can threaten continuity and fail to realise the benefits of supportive relationships (Coady, 2014; Ruch, 2005; Ruch, Turney, & Ward, 2010). This growing interest, and its impact on policy, has extended to residential and foster care (Scottish Government, 2013). Although, this has not yet resulted in consistent application of such principles to work with care leavers (Adley & Jupp Kina, 2014). One reason for this may be that the prevailing practice model for care leavers has been one of 'referring on' from children's services to transition services, rather than promoting a continuity of approach and of relationships (McGhee, Lerpiniere, Welch, Graham, & Harkin, 2014). Nonetheless, the Care Inquiry challenged the assumption that 'old relationships need to be broken in order for new ones to be made', and sought to explore ways of 'ensuring that the benefits of a care option for a particular child will endure beyond childhood' (Care Inquiry, 2013, p. 8).

In making the case for a renewed emphasis on relationship-based practice, many authors recognise that universalised systems of assessment and review can serve to deemphasise the relational aspects of working with a child in care, leading to situations where children have been denied the close relationships that they need (Holland, 2009a; Steckley & Smith, 2011). However, as previously mentioned, forming and maintaining these relationships can itself be challenging. Studies have highlighted the complexity of building and sustaining meaningful and supportive relationships with children and young people, particularly when there is a high likelihood that they have experienced adverse childhood experiences and significant levels of abuse and neglect (Schofield, Beek, & Ward, 2012).

Other policy drivers may, in some ways, clash with those that promote relationshipbased practice. For example, while generally welcoming collaborative approaches promoted by corporate parenting, some commentators have raised concerns that collaboration may de-emphasise relational continuity if this leads to a range of professionals being involved in the life of a young person (Holland, 2009b; Steckley & Smith, 2011). These commentators argue that sometimes no-one takes individual responsibility for following through with specific issues, or indeed offering a closer, more reciprocal relationship to that young person.

Part 3. Themes around relationships

The importance of building networks of social support

Studies have recognised the importance of informal, as well as formal, networks of support, and have pointed to the importance of helping young people to build 'social capital' to reduce dependence on intervention by statutory services (Berridge & Cleaver, 1987; Newman & Blackburn, 2002). In the findings of the *Someone to Care* report, it was highlighted that leaving care support workers with large caseloads of up to 40 young people, could not support vulnerable care leavers (Barnardo's, 2014). The report suggests that both informal networks and other supporting professionals, such as advocacy workers, tenancy support workers, participation workers, and employment advisors, played a key role in meeting young peoples' needs (Barnardo's, 2014). Wade

(2008) further emphasises the importance of establishing informal networks of support, including members of young peoples' birth families, on-going relationships with substitute caregivers, and new families young people attempted to create soon after leaving care (such as those centred on their relationships with partners and their own birth children).

Many young care leavers report feeling lonely, isolated and vulnerable (Duncalf, 2010; Morgan & Lindsay, 2006) and tend to have smaller support networks than the wider population and those still in care (McMahon & Curtin, 2013). Social workers and leaving care support workers are often poorly informed about young peoples' friendships and social networks (Biehal et al., 1995; Marsh & Peel, 1999), and insufficient time is spent looking at this aspect of leaving care (Courtney et al., 2007). As a result, Scannapieco and colleagues explored a model of practice that focussed on creating a support network for young people based on a thorough assessment of their needs, but emphasised the importance of regularly reviewing and updating this, as care leavers' experiences and needs evolve (Scannapieco, Connell-Carrick, & Painter, 2007).

Mentoring programmes have become an increasing feature of the landscape for children on the edge of care, for those in care, and for care leavers. A young person's mentor may fulfil the role of a trusted adult from outside the family, and some studies suggest this can be helpful in building resilience and improving outcomes (Stein, 2007). The role of 'natural mentors' (informal, unpaid supporters) in the lives of older young people in transition from foster care was explored in a study by Greeson and colleagues, who concluded that these relationships had much to offer in alleviating some of the challenges faced by young people (Greeson, Thompson, Evans-Chase, & Ali, 2014). The authors also highlighted the advantages that this informal approach had over formal programmes that can be difficult to establish with older young people.

In their systematic review of mentoring studies, Thompson, Greeson, and Brunsink (2016) argue that natural mentoring relationships may improve outcomes for young people during their transition from care to independent adulthood. Additionally, the Glasgow-based MCR Pathways mentoring scheme, targeting mainly looked after young people of secondary school age, is yielding encouraging results, and has now been extended to more schools following a successful pilot involving six schools (Fassetta, Siebelt, & Mitchell, 2014).

Issues around maintaining relationships and re-connecting with birth families

In contrast to many of their peers, care leavers can often call on little, if any, direct family or support networks within their community (Cashmore & Paxman, 2006). However, some studies have shown that the majority of young people are in touch with their families at the point of leaving care (Cleaver, 2000; Courtney et al., 2007; Farmer, Moyers, & Lipscombe, 2001; Wade, 2008). In examining the findings of three Australian studies, Mendes and colleagues highlight that:

[...] young people who are able to establish positive relationships with their family in care and/or when transitioning from care are more likely to have a positive self-identity and self-confidence, and overall better outcomes. Conversely, poor or non-existent family links may contribute to low self-esteem and limited confidence (Mendes, Johnson, & Moslehuddin, 2012, p. 357).

Many care leavers have experienced poor or disrupted family relationships, or have faced neglect, emotional, physical, or sexual abuse, or rejection (Wade, 2008). As a result, there is no suggestion that informal support provided by the extended family is likely, in itself, to provide sufficient or appropriate support to young people in transition (Coyle & Pinkerton, 2012). Early life trauma can leave care leavers in a position where they find it difficult to form relationships, and to trust adults, particularly where they have had poor relationships with their birth parents (Biehal et al., 1995). In many ways, those young people who are in most need of supportive relationships in their journey to adulthood are often those least able to make and sustain them (Downes, 1992; Stein & Carey, 1986).

Wade's study found that the degree of contact that young people maintain with their families while they are looked after was a good indicator of the level of support that is likely to be available after leaving care. He did, however, find evidence that:

[...] problematic relationships could still evolve to the point where they were supportive, even if the young person would not contemplate living with that person (2008, p. 45).

Other studies (Holland & Crowley, 2013) highlight the importance of helping to develop relationships with siblings and the need for social workers and leaving care services to mediate, working on family issues at the leaving care stage. Although, some studies suggest that professionals do not generally see this as a priority (Biehal et al., 1995). One of the findings of Marsh and Peels' study on the role of extended families in the leaving care process, shows that social workers are not good at identifying these potential supports, or inviting family members into the care planning process (Marsh & Peel, 1999). The leaving care regulations and guidance in England and Wales emphasises that pathway planning should explore all potential sources of support in a young person's kinship network (Department of Health, 2001). Irrespective of whether parents can provide support, 'a young person's wider kin may be able to make a helpful contribution' (Wade, 2008, p. 48). Additionally, care leavers have regularly reported that they wanted to have more contact with their wider family members, and would like to have been better informed about their birth family and the reasons why they became looked after (Coram Voice, 2015; Holland & Crowley, 2013; Wade, 2008).

The importance of preparation, emotional support and interpersonal relationships

Relevant legislation highlights the principle that care leavers should expect the same level of care and support that others would expect from a reasonable parent (Scottish Government, 2008). The Care Inquiry recommended that before and during transitions,

service providers increase their focus on psychological and emotional needs, and building social and relationship networks, as well as proving practical support (Care Inquiry, 2013). However, the literature around leaving care repeatedly draws attention to the insufficient importance attached to this complex aspect of transitions, as compared to more tangible elements such as housing, life skills, and financial support. For example, a 2008 report by the Centre for Social Justice highlights that, in one particular study, '71% of leaving care and personal advisors felt that insufficient attention is paid to emotional support for young care leavers' (Centre for Social Justice, 2008, p. 165). The same report found that a sense of loneliness, isolation, and lack of support was one of the most frequently recurring themes of that consultation. Similarly, in a study by Singer and colleagues, none of the young people surveyed felt that anyone had worked with them to explore or identify gaps in their emotional support network, although they did recognise that some work had been done in introducing them to community supports or a mentor (Singer, Berzin, & Hokanson, 2013). In a Swedish study, Höjer and Sjöblom carried out research with 65 young care leavers and found that the young people wanted emotional support when transitioning from care:

Not only when they felt lonely and vulnerable; many of them said that they needed someone to talk to about their former and present lives and would have liked contact with social workers and former carers for this purpose (2014, p. 2).

When planning and delivering emotional support, providers should remember that early attachment experiences and pre-care relationships might affect the young person's ability, or desire, to engage with support (Adley & Jupp Kina, 2014; Howe, 1995). It is also important to acknowledge that many young people moving on from care turn down offers of support, and that significant numbers of young people become 'switched off to anything related to the care system at the point of leaving care' (Adley & Jupp Kina, 2014, p. 5). The suggestion here is that it is important to find ways to de-stigmatise aftercare support, and to persist with offers well after the young person has moved from placement.

It is argued that a significant number of young care leavers derive emotional and practical support from co-habiting with a partner and, in some case their partner's families. There are a disproportionate number of care leavers who become parents at an early age, which often happens soon after leaving care (Meltzer, Lader, Corbin, Goodman, & Ford, 2004). In reality, however, stability in co-habiting or new parent relationships can be elusive, and they are often transitory or characterised by violence (Wade, 2008).

The centrality of relationships in building resilience and improving outcomes

The concept of resilience features strongly in the literature around looked after children and care leavers.

Resilience can be defined as the quality that enables some young people to find fulfilment in their lives despite their disadvantaged backgrounds, the problems or adversity they may have undergone, or the pressures they may experience [...] it is about overcoming the odds, coping and recovery (Stein, 2007, p. 36)

The resilience of young people from very disadvantaged family backgrounds has been associated with a redeeming, unconditionally supportive, and warm relationship with at least one person, whether that be a parent, other member of family, or parent substitute (Masten, 2009; Rutter, 2000). Having or acquiring resilience helps vulnerable young people and those leaving care to overcome adversity, and obtain better outcomes (Newman & Blackburn, 2002).

Studies suggest that stability, continuity, and secure relationships are key building blocks for resilience, but these are not always available to young people moving on from care, particularly for those whose in-care experiences have been characterised by multiple placements and disruption (Jackson & Martin, 1998; Wade & Munro, 2008). The concept of `felt security' provides a useful way to understand this (Cashmore & Paxman, 2006). Felt security is about young people feeling secure and stable in their care setting; this, in turn, is associated with the development of meaningful and trusting relationships with carers and others who occupy an important place in the child's life (Gilligan, 2008).

Resilience also encompasses the young people's positive sense of identity and personal history, with several authors stressing that those entrusted with care need to help young people to understand why they are in care, and address feelings of rejection and resentment (Biehal et al., 1995; Stein, 2007). In her 2011 paper, Ward acknowledges that young people need a sense of belonging and connectedness if they are to make a successful transition from care. She points out that developing this can be problematic, given that 'care leavers' previous experiences may have made it difficult to establish the secure attachments, sense of stability and sense of self-worth that lie behind perceptions of belonging' (2011, p. 2512). For some young people, a positive journey through care has compensated for earlier adversities, giving them an opportunity to develop their potential in a new 'family' (foster or residential), and accumulate resilience-promoting factors (Newman & Blackburn, 2002; Rutter, Giller, & Hagell, 1998). Ultimately, both strong social support networks and committed mentors or people outside of the family have been highlighted as important in encouraging resilience during transition periods (Hiles, Moss, Wright, & Dallos, 2013).

Part 4. Relationships in and beyond the care setting

In her 2010 study, Duncalf cites a care leaver who provides a useful insight:

If a young person who lives with parents leaves home, they often do so in the knowledge that they can return or at least visit regularly for parents to sort out problems, whether it's just domestic, financial or emotional. I think this is the main difference for care leavers. When they leave care, whether it's a children's home or foster home, they have no feelings that

they have a natural right to return. They go from childhood to the adult world very abruptly (female care leaver, 61, Kent) (p. 29).

As previously stated, young people leaving care face accelerated transitions into adulthood (Biehal et al., 1995). These transitions take a number of forms, depending on the setting from which the young person moves on. Additionally, the maintenance of post-care relationships varies between foster care, residential care and social work settings. As a result, these will be discussed in turn.

Foster care

In 2016, 35% of looked after children and young people were living with foster carers. The majority of these young people (25%) lived with local authority foster carers, whereas the remaining 10% lived with foster carers employed with voluntary organisations and purchased by the local authority (Scottish Government, 2017). Portrayals of foster care suggest it offers children and young people the possibility of an alternative 'family' and 'parental' figures, with the potential that these relationships grow and develop to underpin a settled placement, where relationships endure over time and continue beyond the care setting. Central to this is the concept of a compensatory and redeeming relationship (Jackson, 2002; Wade & Munro, 2008).

Studies have shown, however, that achieving stability in foster care is not straightforward. This is especially true when considered within the context of difficult precare experiences, disrupted relationships, the impact of multiple placements, and possibly, repeated attempts at rehabilitation with birth parents (Stein, 2004). The elusiveness of a secure, stable foster placement is particularly problematic for young people who are placed during late childhood (Jackson & Thomas, 1999).

The nature of the relationship, and of the matching process, is complex and dependant on a number of factors related to the individual child, the fostering family, and the wider support available (Schofield, Beek, Ward, & Biggart, 2013). The relationships between foster carers and the young person moving on from their care must be considered, not least because the role of foster carer straddles the personal and professional domains identified in social pedagogy, and can also extend into the private domain (Holthoff & Eichsteller, 2009). Where carers manage this balancing act well, they are able to move flexibly between the role of professional carers demonstrating codified skills, and the role of parents that may include elements that are tacit and intuitive, often informed by their own parenting experience (Schofield et al., 2012).

Since the 1980's, there has been a strong focus on seeking stability and permanence for children and young people in state care. Often long-term fostering, as well as adoption, is promoted as a valuable opportunity for permanence (Schofield et al., 2012). The objective of permanence, as stated in care planning guidance and regulation is to achieve 'a stable living situation for a child which meets his or her needs for consistent, sustainable, positive relationships, normally best achieved within a family setting' (Scottish Government, 2007, p. 15). This is to ensure that children have a secure and

stable family to support them throughout and beyond their childhood. In line with this, permanency planning is defined as activities that are 'designed to help children live in families that offer continuity of relationships with nurturing parents or caregivers and the opportunity for life-time relationships' (Maluccio & Fein, 1983, p. 196).

The reality of life for many care leavers does not always match the aspirations of social policy; stable, long-term placements are not easy to achieve. A summary of research in the UK by Sinclair and colleagues found that most foster children experience frequent moves, and that they tend to move on from foster placements before turning 18 (Sinclair et al., 2005). Many young people can feel threatened by the thought of close relationships in foster care, whereby they seek to protect themselves from the pain of repeated breakdowns by not allowing themselves to get too close to people (Downes, 1992; Hiles et al., 2013). Foster carers can struggle to understand this, often seeing the young person as 'difficult' if they find it hard to integrate into family life or reciprocate with close or trusting relationships (Jones & Morris, 2012). Equally, where young people do experience a sense of belonging, identifying with their foster carers, this may not always be reciprocated. The resulting conflict can lead to placement breakdown, and further loss for the young person (Biehal & Wade, 1996).

Where stable, long-term foster care is achieved, the resulting sense of 'normal' family life can be undermined or threatened by systemic or bureaucratic processes such as those intended to assure the quality of placements or improve outcomes. For example, Looked After Children Review processes, and requirements for foster carers to seek permissions from social workers for routine decisions and aspects of care (Schofield et al., 2012). Schofield and colleagues highlighted some of the tensions that could arise between carers and leaving care services, whereby:

[...] leaving care social workers, who were introduced just prior to the child's 16th birthday, saw themselves as the key advocates for the young people to help them into 'independence', which often cut across the young people's own view of their long-term place in the foster family, and their foster carers as parents (2012, p. 251).

Studies examining relational issues between young people and their foster carers have pointed out that many young people value the secure and welcoming environment, and the opportunities of belonging that foster care can bring, particularly where this leads to them feeling like they are part of the family. Young people have indicated a strong desire to be treated the same as other children on placement, and the same as the foster carers' birth children (Clarkson, Dallos, Stedmon, & Hennessy, 2017). They also appreciate when foster carers support their educational progress and have high aspirations for them, including encouragement and help in obtaining employment (Arnau-Sabatés & Gilligan, 2015; Christiansen, Havnen, Havik, & Anderssen, 2013; Coram Voice, 2015; Schofield, Beek, & Sargent, 2000). Clarkson et al. (2017) highlight the importance of talking and laughter in the formation of close relationships with foster carers. They suggest that where young people and their foster carers spend time in open and honest conversation, or engaging in meaningful moments, they are able to appreciate the significance of their bonds. Furthermore, other studies have highlighted the importance of addressing factors other than the carer-child relationship, looking at the child's relationships in a broader sense to include other members of the foster and birth family, friends, school and the wider community (Gilligan, 2001; Sinclair, Baker, Lee, & Gibbs, 2007).

For young people in fostering settings, placement and relationship stability is most likely to be achieved when placing children at a younger age, with siblings, and with the intention for the placement to be long-term (Christiansen et al., 2013; Sinclair et al., 2007). Sinclair and colleagues found that, although just under half of the young people who answered a questionnaire said that they wanted to go on living with their foster carers beyond the age of 18, only one in ten actually did so (Sinclair et al., 2005). There is widespread agreement in the literature, as there is in current policy initiatives, about the benefits of 'staying-on' in a stable placement where possible, undertaking a planned, gradual transition from foster care (Stein, 2004). Wade (2008), for instance, found that there were clear benefits to young people who stayed-on in placement, particularly around engagement with education, training and employment. Nonetheless, he indicates that this option was rarely provided. In his study, almost half of the young people were in regular contact with their foster carers three months after moving on, however, this dropped to 14% at the ten-month follow-up (Wade, 2008, p. 49). This finding is broadly in line with other studies (Biehal et al., 1995; Biehal & Wade, 1996; Courtney et al., 2007). Wade's study also emphasised that much of the support provided to young people moving on from foster care, 'occurs informally, without recompense or formal integration into the pathway planning process' (Wade, 2008, p. 49).

The suggestion from these and other studies is that there should be more concerted efforts to make full use of the potential that foster carers have to provide placements well beyond the age of 18. Their ability to provide support beyond the care setting and maintain important supportive relationships from long-term foster care should be promoted further. Foster carers should receive help to understand the changing nature of these relationships and information about how they can adapt to the support needs of former fostered youth. Providers need to establish imaginative ways of supporting these on-going relationships. This might include paying a retainer, or converting former foster carers to support carers, giving them more recognition and clarity of purpose (Biehal & Wade, 1996; Care Inquiry, 2013; Christiansen et al., 2013; Scottish Government, 2013; Sinclair et al., 2005).

Residential care

Of the 10% of looked after children who are placed in residential care, the majority are placed in local authority small group homes (39%), or in a residential school setting (26%) (Scottish Government, 2017). There has been a tendency in recent years for local

authorities and independent providers to focus on smaller four- to six-bedded units rather than larger group homes. Similarly, many traditional residential schools have adapted to incorporate smaller living units, feeding into a co-located or central hub where they offer specialist education (Connelly & Milligan, 2012; Smith, 2009). These smaller units seem to provide particular opportunities for forging meaningful and supportive relationships. Studies point to the many positive views expressed by young people, and evidence of positive relationships for young people in residential care. Alongside this, these studies highlight the range of barriers and challenges that exist in developing and maintaining such relationships (Cashmore & Paxman, 2006; Connelly & Milligan, 2012; Fowler, 2015).

A significant body of research also highlights that young people leaving residential care struggle to achieve successful outcomes. These outcomes are often poor in comparison to the general population, and those leaving other forms of care (Dixon & Stein, 2005; Dumaret, Donati, & Crost, 2011). Nevertheless, it is noted that outcomes for those young people who have experienced residential schooling are markedly better (Martin & Jackson, 2002).

Some commentators have argued that compared to those in foster care or kinship care settings, the cohort of young people placed in residential settings are more likely to have behavioural problems, socialisation difficulties, and to be in contact with mental health services (Gaskell, 2010; Meltzer et al., 2004). This is likely to indicate that young people who are cared for in residential settings have experienced a higher level of deprivation, trauma, and abuse than other looked after children. These early experiences may make it more difficult to benefit from positive aspects of being looked after. Stein uses the term the 'victims group' to describe 'those young people whose experience in care is unlikely to compensate them for early life adversity and disadvantage' (2006, p. 277).

However, we would stress that residential care will be the best response to some young people's needs, and many young people themselves view it positively. In Duncalf's 2010 study, a higher proportion of respondents reported a mainly positive experience of residential care (38%) than those reporting on foster care (26.7%) (Duncalf, 2010, p. 14). Other studies have also found that some young people clearly express a preference for group care as compared to family-based settings (Hill, 2009).

Notwithstanding the difficulties associated with defining 'family', modern residential child care does not portray itself as a family, allowing a child's birth family to be acknowledged and respected (Smith, 2009). Even so, this form of residential care is intended to be 'family like' in that it can provide children with a secure, nurturing, and stimulating environment where they experience warm, authentic relationships with residential workers (Connelly & Milligan, 2012). Additionally, it is notable that in some of the more positive accounts of young people's residential experiences, descriptions often suggest the care was like a 'kind of family' (Duncalf, 2010; Happer, McCreadie, & Aldgate, 2006).

Residential care can protect children from some of the complexities and dangers of family life, particularly when that family life has been marked by extreme stress, dysfunction or abuse (Bolger & Millar, 2012). As Connelly and Milligan suggest:

There are advantages to the group setting where children have the emotional and physical space to develop healing relationships without feeling trapped in a family that is attempting to replace their own (2012, p. 103).

Relationships in residential care can be enhanced by young people's perception that workers are 'there for them', and caring, nurturing relationships are developed by residential staff living through the 'highs and lows' of young people's lives (Houston, 2010, p. 367). Further, research consistently points to the importance of stability of the placement setting and, associated with this, the importance for young people of positive and stable relationships with professionals involved in their care, particularly relationships with their key-workers (Dumaret et al., 2011; Holt & Kirwan, 2012; Martin & Jackson, 2002; McLeod, 2010).

Nonetheless, studies point to particular challenges in building and sustaining relationships with young people in residential care. One challenge relates to the stigmatisation of residential care and the widespread perception that this form of care is the 'option of last resort', only suitable for those that have struggled to manage at home or in foster care settings (Connelly & Milligan, 2012; Kendrick, Milligan, & Avan, 2005; Smith, 2009). Further issues can be identified in the bureaucratic, managerial aspects of residential care, whereby structural barriers prevent young people and staff members from developing strong, protective relationships (Moore et al., 2018).

Other challenges may relate to the nature of the lifespace itself. Commentators highlight the challenges of developing genuine concern and deep level relationships while maintaining 'appropriate' boundaries between the key-worker and the young person, or providing stability when young people have often experienced multiple placement moves, disruption to education, and a series of fractured relationships (Coyle & Pinkerton, 2012). Similarly, a high incidence of staff absence and staff turnaround can cause difficulties in the formation and maintenance of relationships in residential care (Holt & Kirwan, 2012). Another important factor is that young people in residential settings typically enter care at an older age, compared to children in other settings. This can limit their ability and motivation to form close relationships (Smith, 2009). Despite barriers to forming relationships, most young people report being able to bond with at least one adult who they valued highlight (Moore et al., 2018).

The literature suggests that leaving care at an older age can be beneficial to the life chances of young people (Dixon & Stein, 2005; Scotland's Commissioner for Children and Young People, 2009). However, this is a particular challenge for residential settings where a strong 'cultural norm' that prescribes independent living at age 16 persists among both staff members and young people. This norm persists despite the

overwhelming evidence of its damaging effects on the life chances of young people (McGhee et al., 2014; Scotland's Commissioner for Children and Young People, 2009; Scottish Government, 2013; Stein, 2006), and despite the age of young people leaving home in the wider population steadily increasing in recent years (Reid, 2007). There are also the challenges of budgetary pressures, and regulatory issues linked to young adults living alongside younger children (McGhee, 2016).

The intimacy of the 'lifespace' in residential settings may make close relationships inevitable, and reciprocal relationships that arise in close, residential settings can be therapeutic, allowing young people to grow and develop (Ruch, 2005; Smith, 2009; Ward, 2011). The development of close and trusting relationships with young people can be seen as a core professional value for residential care, and social work more broadly (Ruch, 2005; Smith, 2009).

It is important to acknowledge the need to safeguard children from abuse while in residential care settings. This issue has a high profile given the publicity generated by reviews of abuse in residential settings within the UK, such as the Edinburgh Inquiry (Marshall, Jamieson, & Finlayson, 1999) and the Kent report (Kent, 1997). Whilst some have used this as a further justification for managerial control and regulation, others argue that, in addition to a robust regulation and inspection regime, the most important safeguards for children are the culture and relationships within a unit – these should be positive and open, with quality of care and relationships at the centre (Furnivall, 2011; Kendrick, 2013). In terms of good practice:

This should also be marked by the willingness of staff to listen carefully to children, to challenge one another and to facilitate young people's access to independent advocacy (Connelly & Milligan, 2012).

In exploring what young people need from their in-care relationships, a number of studies have found young people value an appropriate relaxation of boundaries within an overall structure. For example, this might include physical contact, sharing personal information, developing special relationships, and offering additional or flexible contact outside of normal working hours (Coady, 2014; Davis & Cree, 2006; Happer et al., 2006; Richmond, 2010). There is also a suggestion that these features can assist young people in developing a sense of belonging and, in many cases, enhancing the feeling that they are indeed part of a 'kind of family' (Kendrick, 2013). They may also help in developing relationships that extend beyond the care setting, particularly in extending the keyworker to young person relationship:

Continuity and stability of relationships that have developed in care but have the capacity to endure post care can continue to offer the young person the secure base from which the transition to independence can occur in a safe and supportive manner (Holt & Kirwan, 2012, p. 377).

It is, therefore, important to recognise that the capacity of the residential unit and the young person to develop, maintain, and sustain meaningful relationships can be critical,

not only to address the effects of earlier problematic attachments, but also as a vital building block in the young person's capacity for independent living (Holt & Kirwan, 2012).

Some writers argue that this capacity is as much about the emotional atmosphere and leadership styles in the unit, as it is about the capacity of individual care staff (Andersson, 2005; Christiansen et al., 2013; Coady, 2014; Holt & Kirwan, 2012). The cultural context is crucial; appropriate boundaries derive more from staff cultures, and the moral stance adopted by individual carers than from particular codes or procedures (Smith, 2009). Some authors have argued that a fundamental rethink of the ethics and practice of residential care is needed in order to reclaim relational practice as a cornerstone of the provision (Steckley & Smith, 2011). A number of authors have pointed to social pedagogy as a framework that lends itself to relationship-based practice, to the use of 'the self' in a caring capacity, and to the therapeutic potential that exists within group care settings (Holthoff & Eichsteller, 2009; Smith, 2009).

Wade and other commentators, particularly young people themselves, have drawn attention to the importance to young care leavers of returning to their previous residential care settings for visits, or to mark special occasions, helping to mitigate social isolation (Broad, 2005; Wade, 2008). In their 2005 study, Dixon and Stein found a third of Scottish local authorities reported having formal policies in relation to providing a continuing role for foster and residential carers beyond the care setting. This included the use of residential units as designated contact points for care leavers, and the use of residential workers to provide continuing outreach work (Dixon & Stein, 2005). A further fifth of local authorities reported that such arrangements could be made on an individual case-to-case basis if it was felt to benefit the young person. Thus, over half of Scottish local authorities recognise Holthoff and Eichsteller's (2009) value of promoting or facilitating on-going relationships with residential carers beyond the time at which a young person leaves the setting.

Studies have pointed out the importance of acknowledging the young person's agency and choice about whether or not to maintain contact, and noted that sometimes relationships may be strained at the point of leaving care (Holt & Kirwan, 2012). A further barrier to continued contact was the subsequent responsibilities and capacity of foster carers and residential workers, as they take on the care of other looked after young people (Dixon & Stein, 2005).

The social work relationship

This review has focussed on the key relationships that develop between young people and their carers within fostering and residential settings, but the literature highlights the important role that other relationships have in supporting young people through the transition from care. For example, research indicates that children and young people value their relationship with their social worker (Larkins et al., 2015). The literature identifies a range of qualities that help young people to form good relationships with their social workers that are also likely to be important in other relationships. Young people value workers who are reliable, honest, available, interested, and effective listeners; they appreciate workers who take them and their views seriously, accept and respect them, are ambitious for them, and who are committed to them (Care Inquiry, 2013; Coram Voice, 2015; Larkins et al., 2015; McLeod, 2010; Morgan & Lindsay, 2006; Ridley et al., 2016; Winter, 2015). In addition, it is very common for the social worker to be the only link between the family background of the young person and their life in care (Winter, 2015).

The quality of these relationships also depends on the length of time the child or young person has known the social worker (Schofield & Stevenson, 2009). In one study seeking the views of 16 care-experienced young people, almost none had had a continuous positive relationship with an adult from their early or mid-childhood (Holland, Floris, Crowley, & Renold, 2010). In a world of restructuring, frequent moves and changes of social worker, it is increasingly unlikely that a single social worker will stay with a child throughout, and beyond, their care journey (Winter, 2015). This is clearly important for young people in understanding their past and in forging their adult identity. Where strong, trusting, enduring relationships exist between social workers and young people, these can be transformational (Who Cares Trust, 2012).

The literature highlights that the interface between the young person's social worker and any leaving care worker is critical to providing effective support for care leavers. There is a variety of service provision arrangements in Scotland for providing these supports. These include: specialist leaving care teams; the extension of children's social work services to care leavers, where the social worker as the key person in co-ordinating support; contracting out support services to the third sector, and; amalgamating services such as leaving care and youth justice services (McGhee et al., 2014). In some instances, residential care staff retain the key role in supporting young people beyond the care setting. Ultimately, Stein and Dixon's study identified that:

Support from social workers and, to some extent, specialist leaving care workers tended to fall away in the early months after leaving care, particularly for young people remaining at home who were previously on a supervision order, or those returning home from a care placement. Where contact had been maintained, help from leaving care workers was generally viewed more positively than help from area social workers, although there was some evidence that young people were confused about who was, or should be, working with them' (Dixon & Stein, 2005, p. 116).

Summary and conclusion

Studies from the UK and other comparable countries clearly indicate the range of challenges and potential hazards that young care leavers can encounter on their journey towards adulthood and independence. They also clearly emphasise the value of supportive relationships in helping young people to overcome these challenges. Notably,

they highlight a number of important issues that need further consideration. These issues are discussed below.

Agency and autonomy of care leavers

In building supportive relationships and extending these relationships beyond the care setting, it is important to recognise the agency of the young person. In the testing circumstances of leaving care, young people will want, and need, to assert their autonomy in ways that their circumstances may have prevented in the past. Research involving young people shows their own concerns about what happens when they are not offered effective support, or when they reject or disengage with support, and their indication that mentors in particular should remain contactable for advice at any point in the future (Newton et al., 2017). It is clear that the rationale for recent policy changes in Scotland is to offer care leavers coherent support through their transition to adulthood and to enable this support to continue for longer - this must include consideration of the continuity of relationships. Young people, whose early lives have often been characterised by the need to survive in an uncertain world, may find it hard to make and sustain relationships and to make use of support. Considerable creativity, as well as persistence, will be required on the part of support services to encourage young people's engagement or re-engagement, ensuring that relationships are central to this.

Supportive relationships

We do, however, know much about what works to achieve these continued relationships. There is a clear, consistent message from research that, where supportive relationships are in place, they are central to the process of successful transitions for young care leavers. We should not leave these relationships to chance. The literature identifies the important role of managers and practitioners in facilitating these key relationships, particularly where young people lack family or community-based support networks (Moore et al., 2018). This type of support can be crucial for young care leavers, particularly those that leave care as young as aged 16 or 17. In order to realise this potential, it may be necessary to challenge some of the attitudes and service structures that run counter to the continued provision of support and relationships. Such provision includes leaving care services that take over from previous supports with little overlap or scope for joint working.

Informal networks of support

The literature also highlights the importance of helping young people to build informal networks of support to complement statutory and professional provisions. One strong theme that emerges is the considerable potential in harnessing the power and commitment of existing relationships, and facilitating the extension of these beyond the care setting, as a formal, or semi-formal, aspect of a young person's support plan (Marion et al., 2017). The importance of supporting young people to develop and build resilience does not stop when they leave care, and the need for a committed and supportive adult in their lives is often stronger than ever at this point. This support should help young people to avoid or reduce feelings of loneliness or isolation.

Emotional support

It is easy to overlook the importance of emotional support in favour of practical, taskoriented work in preparing young people to leave care and supporting them through the transition. Studies suggest that supporting agencies should devote more time and resources to exploring the reasons why young people were looked after. They should take time to reflect with them on the implications of leaving care on their relationships with their birth family, and on their interpersonal relationships more generally, particularly where young people have faced adverse childhood experiences (Smith, 2009). The indications are that in approaching this transition more pro-actively, care leaving services may be pre-empting problems, benefitting care leavers by increasing their self-awareness and problem-solving skills. Given that the majority of care leavers are likely to have substantial contact with their family or extended family on leaving care (Wade & Munro, 2008), it would appear to be important for services to acknowledge and help young people to deal with the consequences of this.

Supporting former carers

In highlighting the potential of continuing relationships beyond care, the research notes the need to provide structure and support to former carers if they are to continue assisting young people. The evidence suggests that if these processes are left to chance, contact is unlikely to be maintained, particularly if there was conflict in the latter stages of the placement. There are clearly significant numbers of relationships that are sustained well beyond the ending of the placement, often enduring for as long as they are needed; it seems those young people have benefitted greatly as a result. Where placements end because of conflict, this often goes unresolved, ultimately leading to a situation where restarting contact without support is not a realistic option for young people.

Relationship-based practice

In the case of residential care, there is a growing interest in reclaiming the central importance of relationship-based practice, but considerable barriers need to be tackled before support beyond the care setting becomes core practice (McGhee, 2016). These include training and awareness-raising for staff and young people in residential units, as well as clear, consistent guidance from managers about the importance of relationships in and beyond the care setting. Taking this action could be central in developing a smooth transition for young people moving on from residential care, ensuring positive and supportive relationships with adults involved in their care continue to be maintained and encouraged.

Supportive environment for practitioners

If the 'relationship is the intervention', the uncertainty of practitioners to engage fully in that relationship can only be detrimental to young people (Fewster, 2004). Attachment-informed and relationship-based practice needs to be at the heart of any good service (Care Inquiry, 2013; Furnivall, 2011; McGhee, 2016; Winter, 2015). Creating a model of care underpinned by this approach requires staff to be encouraged and empowered to

develop their practice within a supportive culture (Scottish Government, 2013; Trevithick, 2014). The impact of the *Staying Put Scotland* initiative (Scottish Government, 2013) has the potential to provide an enabling context for these practices and the relational elements of this. The impact of these recent policy and legislative changes may not become fully clear for some time.

Empowering care leaver's

The steady increase in advocacy organisations and care leaver's participation groups suggest there is a growing group of empowered young care leavers who wish to speak out and influence services. As Duncalf concluded in her study:

There needs to be recognition that care leavers of all ages have a wealth of knowledge and experience that can benefit current practices and practitioners. In light of this, there needs to be greater involvement of care leavers of all ages in research, policy and practice (2010, p. 42).

In conducting this review, there is strong evidence to support the Care Inquiry's conclusions that:

High-quality relationships matter more than anything else for children in or on the edge of care [...] and we need a care system that places at its heart the quality and continuity of relationships, and that promotes and enhances the ability of those who are important to children – care givers and others – to provide the care and support they need (2013, p. 8).

As a final note, we offer a concluding quote from Carrie Reid's paper reflecting on the situation in Canada:

Research shows that youth who maintain relationships with at least one supportive adult are far more likely to go on to have successful outcomes. Thus, a key factor for the success of youth leaving care is the same as what all other youth need: an adult who cares and provides support in good times and bad (2007, p. 36).

References

- Adley, N., & Jupp Kina, V. (2014). Getting behind the closed door of care leavers: Understanding the role of emotional support for young people leaving care. *Child & Family Social Work*.
- Aldgate, J. (1994). Graduating from care: A missed opportunity for encouraging successful citizenship. *Children* and Youth Services Review, 16(3), 255-272.
- Andersson, G. (2005). Family relations, adjustment and well-being in a longitudinal study of children in care. *Child & Family Social Work, 10*(1), 43-56.
- Arnau-Sabatés, L., & Gilligan, R. (2015). What helps young care leavers to enter the world of work? Possible lessons learned from an exploratory study in Ireland and Catalonia. *Children and Youth Services Review*, 53, 185-191.
- Barnardo's. (2014). Someone to care: Experiences of leaving care. Ilford: Barnardo's.
- Berridge, D., & Cleaver, H. (1987). Foster home breakdown. Oxford: Blackwell.
- Biehal, N., Clayden, J., Stein, M., & Wade, J. (1995). *Moving on: Young people and leaving care schemes*. London: HMSO.
- Biehal, N., & Wade, J. (1996). Looking back, looking forward: Care leavers, families and change. *Children and Youth Services Review*, *18*(4), 425-445.
- Bolger, J., & Millar, J. (2012). Residential child care in practice. In M. Davies (Ed.), *Social Work with Children and Families* (pp. 304-322). Basingstoke: Palgrave Macmillan.
- Broad, B. (1998). *Young people leaving care: Life after the Children Act 1989*. London: Jessica Kingsley Publishers.
- Broad, B. (2005). *Improving the health and well-being of young people leaving care*. Lyme Regis: Russell House Press.
- Broad, B. (2007). Care leavers in transition: DfES youth strategy review. Leicester: De Montfort University.
- Bynner, J., Elias, P., McNight, A., Pan, H., & Pierre, G. (2002). *Young people's changing routes to independence*. York: Joseph Rowntree Foundation.
- Care Inquiry. (2013). *Making not breaking: Building relationships for our most vulnerable children*. London: House of Commons.
- Cashmore, J., & Paxman, M. (2006). Predicting after-care outcomes: The importance of 'felt'security. *Child & Family Social Work*, *11*(3), 232-241.
- Centre for Social Justice. (2008). *Couldn't care less: A policy report from the Children in Care working group*. London: Centre for Social Justice.
- Christiansen, Ø., Havnen, K. J., Havik, T., & Anderssen, N. (2013). Cautious belonging: Relationships in long-term foster-care. *British Journal of Social Work, 43*(4), 720-738.
- Clarkson, H., Dallos, R., Stedmon, J., & Hennessy, C. (2017). Exploring the relationship: Joint narratives of foster carers and young people. *Adoption and Fostering*, *41*(1), 35-51. doi:10.1177/0308575916681711

- Cleaver, H. (2000). Fostering family contact: A study of children, parents and foster carers. *British Journal of Psychotherapy*, *12*, 195-207.
- Coady, P. (2014). Relationship boundaries in residential child care: Connection and safety in group care relationships. *Research, Policy and Planning, 31*(2), 79-91.
- Connelly, G., & Milligan, I. (2012). *Residential child care: Between home and family*. Edinburgh: Dunedin Academic Press.
- Coram Voice. (2015). *Children and young people's views on being in care: A literature review*. London: Coram Voice.
- Courtney, M. E., Dworsky, A. L., Cusick, G. R., Havlicek, J., Perez, A., & Keller, T. E. (2007). *Midwest evaluation of the adult functioning of former foster youth: Outcomes at age 21*. Chicago: Chapin Hall Center for Children.
- Coyle, D., & Pinkerton, J. (2012). Leaving care: The need to make connections. *Child Care in Practice, 18*(4), 297-308.
- Davis, A., & Cree, V. (2006). Social work: Voices from the inside. London: Routledge.
- Department for Education, Cabinet Office, Department for Communities and Local Government, Department for Work and Pensions, Department of Health, HM Treasury; Home Office, & Ministry of Justice (2016). *Keep on caring: supporting young people from care to independence*. London: HM Government.
- Department of Health. (2001). *Children (Leaving Care) Act 2000: Regulations and guidance*. London: Department of Health.
- Dixon, J., & Stein, M. (2005). Throughcare and aftercare in Scotland. London: Jessica Kingsley Publishers.
- Downes, C. (1992). Separation revisited: Adolescents in foster family care. Aldershot: Ashgate.
- Dumaret, A. C., Donati, P., & Crost, M. (2011). After a long-term placement: Investigating educational achievement, behaviour, and transition to independent living. *Children & Society, 25*(3), 215-227.
- Duncalf, Z. (2010). *Listen up! Adult care leavers speak out: The views of 310 care leavers aged 17-78*. Manchester: Care Leavers' Association.
- Farmer, E., Moyers, S., & Lipscombe, J. (2001). The fostering task with adolescents. Bristol: University of Bristol.
- Fassetta, G., Siebelt, L., & Mitchell, I. (2014). *Transforming lives: The first 18 months of the MCR Pathways Mentoring Programme*. Glasgow: Glasgow MCR Pathways / CELCIS.
- Fewster, G. (2004). Just between you and me. Relational Child and Youth Care Practice, 17(4), 8-17.
- Fowler, N. (2015). 'We're like one, big, dysfunctional family': Struggling to define the role of residential child care workers. *Scottish Journal of Residential Child Care*, 14(3), 19-32.
- Furnivall, J. (2011). Attachment-informed practice with looked after children and young people. Glasgow: IRISS.

Gaskell, C. (2010). 'If the social worker had called at least it would show they cared': Young care leaver's perspectives on the importance of care. *Children & Society*, *24*(2), 136-147.

- Gilligan, R. (2001). *Promoting resilience: A resource guide on working with children in the care system*. London: British Agencies for Adoption and Fostering.
- Gilligan, R. (2008). Promoting resilience in young people in long-term care: The relevance of roles and relationships in the domains of recreation and work. *Journal of Social Work Practice, 22*(1), 37-50.
- Greeson, J. K., Thompson, A. E., Evans-Chase, M., & Ali, S. (2014). Child welfare professionals' attitudes and beliefs about child welfare-based natural mentoring for older youth in foster care. *Journal of Social Service Research*, 41(1), 93-112.
- Happer, H., McCreadie, J., & Aldgate, J. (2006). *Celebrating success: What helps looked after children succeed*. Edinburgh: Social Work Inspection Agency.
- Hiles, D., Moss, D., Wright, J., & Dallos, R. (2013). Young people's experience of social support during the process of leaving care: A review of the literature. *Children and Youth Services Review*, 35(12), 2059-2071.
- Hill, M. (2009). *Higher aspirations, brighter futures: Matching resources to needs report*. Glasgow: Scottish Institute for Residential Child Care.
- Höjer, I., & Sjöblom, Y. (2014). Voices of 65 young people leaving care in Sweden: "There is so much I need to know!". *Australian Social Work, 67*(1), 71-87.
- Holland, S. (2009a). Listening to children in care: A review of methodological and theoretical approaches to understanding looked after children's perspectives. *Children & society, 23*(3), 226-235.
- Holland, S. (2009b). Looked after children and the ethic of care. *British Journal of Social Work, 40*(6), 1664-1680.
- Holland, S., & Crowley, A. (2013). Looked-after children and their birth families: using sociology to explore changing relationships, hidden histories and nomadic childhoods. *Child & Family Social Work, 18*(1), 57-66. doi:10.1111/cfs.12032
- Holland, S., Floris, C., Crowley, A., & Renold, E. (2010). 'How was your day?' Learning from experience: informing preventative policies and practice by analysing critical moments in care leavers' life histories. Cardiff: Cardiff University.
- Holt, S., & Kirwan, G. (2012). The "key" to successful transitions for young people leaving residential child care: The role of the keyworker. *Child Care in Practice, 18*(4), 371-392.
- Holthoff, S., & Eichsteller, G. (2009). Social pedagogy: The practice. *Every Child Journal, 1*(1), 58-63.
- Houston, S. (2010). Building resilience in a children's home: Results from an action research project. *Child & Family Social Work, 15*(3), 357-368.
- Howe, D. (1995). Attachment theory for social work practice. London: MacMillan.
- Jackson, S. (2002). Nobody ever told us school mattered: Raising the educational attainments of children in care Response. *British Journal of Educational Psychology, 72*, 141-142.
- Jackson, S., & Martin, P. Y. (1998). Surviving the care system: Education and resilience. *Journal of adolescence*, 21(5), 569-583.

- Jackson, S., & Thomas, N. (1999). On the move again: What works in creating stability for looked after children (Barnardo's Ed.). Ilford: Barnardo's.
- Jones, A. M., & Morris, T. L. (2012). Psychological adjustment of children in foster care: Review and implications for best practice. *Journal of Public Child Welfare, 6*(2), 129-148.
- Kendrick, A. (2013). Relations, relationships and relatedness: Residential child care and the family metaphor. *Child & Family Social Work, 18*(1), 77-86.
- Kendrick, A., Milligan, I., & Avan, G. (2005). Nae too bad': Job satisfaction and staff morale in Scottish residential child care. *Scottish Journal of Residential Child Care*, 4(1), 22.
- Kent, R. (1997). Children's safeguards review. Edinburgh: Scottish Office.
- Larkins, C., Ridley, J., Farrelly, N., Austerberry, H., Bilson, A., Hussein, S., . . . Stanley, N. (2015). Children's, young people's and parents' perspectives on contact: Findings from the evaluation of social work practices. *British Journal of Social Work, 45*(1), 296-312.
- Lee, B. R., Cole, A. R., & Munson, M. R. (2016). Navigating family roles and relationships: system youth in the transition years. *Child and Family Social Work, 21*(4), 442-451. doi:10.1111/cfs.12160
- Maluccio, A. N., & Fein, E. (1983). Permanency planning: A redefinition. Child Welfare, 62(3).
- Marion, É., Paulsen, V., & Goyette, M. (2017). Relationships Matter: Understanding the Role and Impact of Social Networks at the Edge of Transition to Adulthood from Care. *Child and Adolescent Social Work Journal, 34*(6), 573-582. doi:10.1007/s10560-017-0494-4
- Marsh, P., & Peel, M. (1999). *Leaving care in partnership: Family involvement with care leavers*. London: The Stationery Office.
- Marshall, K., Jamieson, C., & Finlayson, A. (1999). *Edinburgh's children: The report of the Edinburgh enquiry into abuse and protection of children in care*. Edinburgh: City of Edinburgh Council.
- Martin, P. Y., & Jackson, S. (2002). Educational success for children in public care: Advice from a group of high achievers. *Child & family social work, 7*(2), 121-130.
- Masten, A. S. (2009). Ordinary magic: Lessons from research on resilience in human development. *Education Canada*, *49*(3), 28-32.
- McGhee, K. (2016). Professional enquiry & development in residential child care: Unpublished.
- McGhee, K., Lerpiniere, J., Welch, V., Graham, P., & Harkin, B. (2014). *Throughcare and aftercare services in Scotland's local authorities: A national study*. Glasgow: CELCIS.
- McLeod, A. (2010). 'A friend and an equal': Do young people in care seek the impossible from their social workers? *British Journal of Social Work, 40*(3), 772-788.
- McMahon, C., & Curtin, C. (2013). The social networks of young people in Ireland with experience of long-term foster care: Some lessons for policy and practice. *Child & Family Social Work, 18*(3), 329-340.
- Meltzer, H., Lader, D., Corbin, T., Goodman, R., & Ford, T. (2004). *The mental health of young people looked after by local authorities in Scotland*. London: The Stationery Office.

- Mendes, P., Johnson, G., & Moslehuddin, B. (2012). Young people transitioning from out-of-home care and relationships with family of origin: An examination of three recent Australian studies. *Child Care in Practice*, *18*(4), 357-370.
- Moore, T., McArthur, M., Death, J., Tilbury, C., & Roche, S. (2018). Sticking with us through it all: The importance of trustworthy relationships for children and young people in residential care. *Children and Youth Services Review*, *84*, 68-75. doi:10.1016/j.childyouth.2017.10.043
- Morgan, R., & Lindsay, M. (2006). Young people's views on leaving care: What young people in, and formerly in, residential and foster care think about leaving care A Children's Rights Director's report. London: Commission for Social Care.
- Newman, T., & Blackburn, S. (2002). *Transitions in the lives of children and young people, resilience factors*. Edinburgh: Scottish Government
- Newton, J. A., Harris, T. O., Hubbard, K., & Craig, T. K. J. (2017). Mentoring During the Transition from Care to Prevent Depression: Care Leavers' Perspectives. *Practice, 29*(5), 317-330. doi:10.1080/09503153.2017.1329414
- O'Leary, P., Tsui, M.-S., & Ruch, G. (2013). The boundaries of the social work relationship revisited: Towards a connected, inclusive and dynamic conceptualisation. *British Journal of Social Work*, 43(1), 135-153.
- Pinkerton, J., & McCrea, R. (1999). *Meeting the challenge? Young people leaving care in Northern Ireland*. Aldershot: Ashgate.
- Reid, C. (2007). The transition from state care to adulthood: International examples of best practices. *New Directions for Youth Development, 2007*(113), 33-49.
- Richmond, P. (2010). Boundary realities from the wisdom of female youth in residential treatment. *Journal of Child and Youth Care Work*, 80-93.
- Ridley, J., Larkins, C., Farrelly, N., Hussein, S., Austerberry, H., Manthorpe, J., & Stanley, N. (2016). Investing in the relationship: Practitioners' relationships with looked-after children and care leavers in Social Work Practices. *Child and Family Social Work, 21*(1), 55-64. doi:10.1111/cfs.12109
- Ruch, G. (2005). Relationship-based practice and reflective practice: Holistic approaches to contemporary child care social work. *Child & Family Social Work, 10*(2), 111-123.
- Ruch, G., Turney, D., & Ward, A. (Eds.). (2010). *Relationship-based social work: Getting to the heart of practice*. London: Jessica Kingsley Publishers.
- Rutter, M. (2000). Children in substitute care: Some conceptual considerations and research implications. *Children and Youth Services Review*, 22(9), 685-703.
- Rutter, M., Giller, H., & Hagell, A. (1998). *Antisocial behavior by young people: A major new review*. Cambridge: Cambridge University Press.
- Scannapieco, M., Connell-Carrick, K., & Painter, K. (2007). In their own words: Challenges facing youth aging out of foster care. *Child and Adolescent Social Work Journal, 24*(5), 423-435.

Schofield, G., Beek, M., & Sargent, K. (2000). Growing up in foster care. London: BAAF.

Schofield, G., Beek, M., & Ward, E. (2012). Part of the family: Planning for permanence in long-term family foster care. *Children and Youth Services Review*, *34*(1), 244-253.

- Schofield, G., Beek, M., Ward, E., & Biggart, L. (2013). Professional foster carer and committed parent: Role conflict and role enrichment at the interface between work and family in long-term foster care. *Child & Family Social Work*, 18(1), 46-56.
- Schofield, G., & Stevenson, O. (2009). Contact and work with birth families. In G. Schofield & O. Stevenson (Eds.), *The child placement handbook: research, policy and practice*. London: BAAF.
- Scotland's Commissioner for Children and Young People. (2009). *Sweet 16? The age of leaving care in Scotland: one year on Is life any sweeter?* Edinburgh: Scotland's Commissioner for Children and Young People.
- Scottish Executive. (2004). Supporting young people leaving care in Scotland: Regulations and guidance on services for young people ceasing to be looked afters by local authorities. Edinburgh: Scottish Executive.
- Scottish Government. (2007). *Permanence planning and decision making for looked after children in Scotland*. Edinburgh: Scottish Government.
- Scottish Government. (2008). *These are our bairns: A guide for community planning partnerships on being a good corporate parent*. Edinburgh: Scottish Government.
- Scottish Government. (2013). *Staying put Scotland: Providing care leavers with connectedness and belonging*. Edinburgh: Scottish Government.
- Scottish Government. (2017). *Children's Social Work Statistics Scotland, 2015-16*. Edinburgh: Scottish Government
- Sinclair, I., Baker, C., Lee, J., & Gibbs, I. (2007). *The pursuit of permanence: A study of the English care system*. London: Jessica Kingsley Publishers.
- Sinclair, I., Baker, C., Wilson, K., & Gibbs, I. (2005). *Foster Children: Where They Go and How They Get On*. London: Jessica Kingsley Publishers.
- Singer, E., Berzin, S., & Hokanson, K. (2013). Voices of fromer foster youth: Supportive relationships in the transition to adulthood. *Children and Youth Services Review*, *35*(12), 2110-2117.
- Smith, M. (2009). Rethinking residential child care: Positive perspectives. Bristol: Policy Press.
- Steckley, L., & Smith, M. (2011). Care ethics in residential care: A different voice. *Ethics and Social Welfare,* 5(2), 181-195.
- Stein, M. (2004). What works for young people leaving care. Ilford: Barnardo's.
- Stein, M. (2006). Research review: Young people leaving care. Child & Family Social Work, 11(3), 273-279.
- Stein, M. (2007). Resilience and young people leaving care. Child Care in Practice, 14(1), 35-44.
- Stein, M. (2012). Young people leaving care. London: Jessica Kingsley Publishers.
- Stein, M., & Carey, K. (1986). Leaving Care. Oxford: Blackwell.
- Sulimani-Aidan, Y. (2017). 'She was like a mother and a father to me': searching for the ideal mentor for youth in care. *Child and Family Social Work, 22*(2), 862-870. doi:10.1111/cfs.12306

- Thompson, A. E., Greeson, J. K. P., & Brunsink, A. M. (2016). Natural mentoring among older youth in and aging out of foster care: A systematic review. *Children and Youth Services Review, 61*, 40-50. doi:10.1016/j.childyouth.2015.12.006
- Trevithick, P. (2014). Humanising managerialism: Reclaiming emotional reasoning, intuition, the relationship, and the knowledge and skills in social work. *Journal of Social Work Practice, 28*(3), 287-311.
- Wade, J. (1997). Developing leaving care services: Tapping the potential of foster carers. *Adoption & Fostering,* 21(3), 40-49.
- Wade, J. (2008). The ties that bind: Support from birth families and substitute families for young people leaving care. *British Journal of Social Work, 38*(1), 39-54.
- Wade, J., & Munro, E. (2008). United Kingdom. In M. Stein & E. Munro (Eds.), Young people's transition from care to adulthood: International research and practice. London: Jessica Kingsley Publishers.
- Ward, H. (2011). Continuities and discontinuities: Issues concerning the establishment of a persistent sense of self amongst care leavers. *Children and Youth Services Review*, 33(12), 2512-2518.
- Who Cares Trust. (2012). *Making it human: Inspirational stories of transformative relationships between social workers and looked-after young people*. London: Who Cares Trust.
- Winter, K. (2015). Supporting positive relationships for children and young people who have experience of care (Vol. 28). Glasgow: IRISS.

About CELCIS

CELCIS, based at the University of Strathclyde in Glasgow, is committed to making positive and lasting improvements in the wellbeing of Scotland's children living in and on the edges of care. Ours is a truly collaborative agenda; we work alongside partners, professionals and systems with responsibility for nurturing our vulnerable children and families. Together we work to understand the issues, build on existing strengths, introduce best possible practice and develop solutions. What's more, to achieve effective, enduring and positive change across the board, we take an innovative, evidence-based improvement approach across complex systems.

For more information

Visit: www.celcis.org Email: celcis@strath.ac.uk Tel: 0141 444 8500